Author’s response to reviews

Title: Early and late mortality after malaria in young children in Papua, Indonesia

Authors:

Dewi Patriani (dewipatriani@yahoo.com)
Eggi Arguni (eggiarguni@gmail.com)
Saber Dini (saber.dini@unimelb.edu.au)
Paulus Sugianto (sugiantopaulus@gmail.com)
Afdhal Hasanuddin (afdalhs_805@yahoo.co.id)
Daniel Lampah (aditimika@yahoo.com)
Nicholas Douglas (nicholas.douglas@gmail.com)
Nicholas Anstey (Nicholas.Anstey@menzies.edu.au)
Julie Simpson (julieas@unimelb.edu.au)
Ric Price (rprice@menzies.edu.au)
Jeanne Rini Poespoprodjo (didot2266@yahoo.com)

Version: 1 Date: 15 Jun 2018

Author’s response to reviews:

Dear Editor,

Thank you for your comments on the Ethics approval and consent to participate section of the manuscript.

The data used in this study is a secondary data from hospital electronic records and therefore individual parental consent was not obtained. Patient’s identification used in this study is hospital record number without name and address. The process was approved by both the ethics review board of the University of Gadjah Mada in Indonesia and Menzies School of Health in Australia.

We have added the statement in line 429 to 433 in the manuscript.

We hope that the manuscript is now ready for review.

Yours sincerely,
Rini